Author's response to reviews

Title: Evaluation of a new disposable silicon limbal relaxing incision knife by experienced users

Authors:

John Albanese (John_Albanese@bd.com)
Geoffrey Dugue (Geoffery_Dugue@bd.com)
Valentin Parvu (Valentine_Parvu@bd.com)
Ann M Bajart (ambajart@eyeboston.com)
Edwin Lee (Edwin_Lee@bd.com)

Version: 4 Date: 14 December 2009

Author's response to reviews: see over
14 December 2009

Dear Editor in Chief,

Enclosed is the manuscript titled: ‘Evaluation of a New Disposable Silicon Limbal Relaxing Incision Knife by Experienced Users’, with the required formatting changes made. Please note all major and minor edits were made including those to the figures.

Thank your for accepting this article for publication.

Sincerely,

John Albanese, MBA

Corresponding Author Information:

John Albanese, MBA
Becton Dickinson and Company
1 Becton Drive, Franklin Lakes, NJ 07417 USA
Ph: 201-847-6764
Fax: 201-847-5213
Email: John_Albanese@bd.com